Author's response to reviews

Title: Breast and Bowel Cancer Screening Uptake Patterns over 15 Years for UK South Asian Ethnic Minority Populations, Corrected for Differences in Socio-demographic Characteristics

Authors:

Ala K Szczepura (ala.szczepura@warwick.ac.uk)
Charlotte L Bean (c.l.bean@warwick.ac.uk)
Anil K Gumber (a.k.gumber@warwick.ac.uk)

Version: 4 Date: 22 September 2008

Author's response to reviews:

Dear Editorial Production Team

As requested, we have uploaded all figures (cropped as requested.

We have left justified all text & have altered the layout of the tables to the required format. We have also made a very minor revision to first paragraph page 10 ("Logistic regression....."); the second sentence has been simplified.

The revised text & tables have been uploaded.